Author’s response to reviews

Title: All-trans retinoic acid stimulates the secretion of TGF-β2 via the phospholipase C but not the adenylyl cyclase signaling pathway in retinal pigment epithelium cells

Authors:
Daren Zhang (zhangdaren@csu.edu.cn)
Shuirong Liu (128167864@qq.com)
Shuyu Hu (hushuyu1995@163.com)
Hui Tao (975598455@qq.com)
Renhong Tang (trh_1@126.com)
Zhihong Deng (daneyldeng@126.com)
Jia Tan (jasmintj@126.com)

Version: 2 Date: 25 Nov 2018

Author’s response to reviews:

Author Reply to comments

BOPH-D-18-00528R1

All-trans retinoic acid stimulates the secretion of TGF-β2 via the phospholipase C but not the adenylyl cyclase signaling pathway in retinal pigment epithelium cells.

Daren Zhang; Shuirong Liu; Shuyu Hu; Hui Tao; Renhong Tang; Zhihong Deng; Jia Tan

BMC Ophthalmology

Dear Dr Deng,

Your manuscript "All-trans retinoic acid stimulates the secretion of TGF-β2 via the phospholipase C but not the adenylyl cyclase signaling pathway in retinal pigment epithelium cells." (BOPH-D-18-00528R1) has been assessed. I am pleased to inform you that it is potentially acceptable for publication in BMC Ophthalmology, once you have carried out some essential revisions
-- We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Introduction/Methods/Discussion sections.

While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.

If there is overlap in the Methods section, please ensure to summarize the methods and cite the source.

Please re-phrase these sections to minimise overlap.

Reply: we have rephrased these sections in the revised manuscript.

-- As you have changed your author list from the order previously on the title page of your manuscript you will need to fill out a change in authorship form and send it to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Reply: We did not change the author list from the order previously on the title page of our manuscript, but we set co-corresponding authors in our paper.

-- Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

Reply: We reuploaded the files.
In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Reply: The funding body of our research is a government organization which supported money only, and it did not make any other contribution to this research.

-- Thank you for providing a Cover Letter. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Reply: We removed it.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We reloaded the revised manuscript as described.

Once you have made the necessary corrections, please submit a revised manuscript online at:

https://boph.editorialmanager.com/

If you have forgotten your password, please use the 'Send Login Details' link on the login page at https://boph.editorialmanager.com/. For security reasons, your password will be reset.

We request that a point-by-point response letter accompanies your revised manuscript. This letter must provide a detailed response to each reviewer/editorial point raised, describing what amendments have been made to the manuscript text and where these can be found (e.g. Methods section, line 12, page 5). If you disagree with any comments raised, please provide a detailed rebuttal to help explain and justify your decision.
Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Submission Guidelines on the journal homepage.

A decision will be made once we have received your revised manuscript, which we expect by 30 Nov 2018.

Please note that you will not be able to add, remove, or change the order of authors once the editor has accepted your manuscript for publication. Any proposed changes to the authorship must be requested during peer-review, and adhere to our criteria for authorship as outlined in BioMed Central's policies. To request a change in authorship, please download the 'Request for change in authorship form' which can be found here - http://www.biomedcentral.com/about/editorialpolicies#authorship. Please note that incomplete forms will be rejected. Your request will be taken into consideration by the editor, and you will be advised whether any changes will be permitted. Please be aware that we may investigate, or ask your institute to investigate, any unauthorized attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

We look forward to receiving your revised manuscript and please do not hesitate to contact us if you have any questions.

Best wishes,
Amelia De Salis
on behalf of,

Amy Branch-Hollis
BMC Ophthalmology

https://bmcophthalmol.biomedcentral.com/
BMC Ophthalmology operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Khrishen Cunnusamy (Reviewer 2): The authors of the manuscript have satisfactorily addressed the questions I had in regards to the manuscript and mention that they will be investigating the ultrastructural effects of inhibiting TGFB2 in subsequent manuscripts. Based on their response and revisions, I deem the manuscript acceptable and suitable for publication in BMC Ophthalmology.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Editorial Policies

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.
In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Ophthalmology must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcophthalmol.biomedcentral.com/).

Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

Further information about our editorial policies can be found at the following links:

Ethical approval and consent:
http://www.biomedcentral.com/about/editorialpolicies#Ethics

Availability of data and materials section:

http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials

Recipients of this email are registered users within the Editorial Manager database for this journal. We will keep your information on file to use in the process of submitting, evaluating and publishing a manuscript. For more information on how we use your personal details please see our privacy policy at https://www.springernature.com/production-privacy-policy or email dataprotection@springernature.com. If you no longer wish to receive messages from this journal or you have questions regarding the Editorial Manager database and the publishing process, please email our publication office, stating the journal name(s) and your email address(es):

PublicationOfficeSPI@springernature.com

In compliance with data protection regulations, please contact the publication office if you would like to have your personal information removed from the database.